\begin{table}%t1
\caption{Parameters used in the different generated SEDs.}
\label{tab:param}
%\centering
\small
\begin{tabular}{l|c|c|cccccc}
  \hline \hline
  Model & Blob-in-jet & Single blob & \multicolumn{6}{c}{Multi-blob}\\
	  &       & in broadened zone & & & & & & \\ 
	  Object & M~87 & M~87 & M~87 & M~87 & 3C~273 & Cen~A & Cen~A & PKS~0521$-$36\\
	  figure & \ref{fig:M87_SED_bij_broad} ({\it green}) & \ref{fig:M87_SED_bij_broad} ({\it blue}) & \ref{fig:M87_SED1} & \ref{fig:M87_SED2} & \ref{fig:3C273_SED} & \ref{fig:CenA_SED} ({\it blue}) & \ref{fig:CenA_SED} ({\it green}) & \ref{fig:pks0521_SED}\\
  \hline
  $\delta_{\rm b}$ & 3.07 & 8 & -- & -- & --& -- & -- & --\\
  $\Gamma_{\rm b}$ & -- & -- & 4.1 & 10.0 & 7.4 & 8.14 & 20.0 & 1.5\\
  $\theta$ & $19\degr$ & $1\degr$ & $15\degr$ & $15\degr$ & $15\degr$ & $25\degr$ & $25\degr$ & $25\degr$\\
  $R_{\rm cap}$ [$r_g$] & -- & -- & 100.0 & 100.0 & 100.0 & 100.0 & 100.0 & 100.0\\
  $B$ [G] & 1.0 & 0.5 & 0.5 & 0.01 & 3.0 & 2.0 & 10.0 & 1.0\\
  $r_{\rm b}$ [cm] & $1.2 \times 10^{14}$ & $4.0 \times 10^{13}$ & $1.5 \times 10^{13}$ & $2.8 \times 10^{14}$ & $2.0 \times 10^{15}$ & $1.0 \times 10^{14}$ & $8.0 \times 10^{13}$ & $9.0 \times 10^{14}$\\
  $K_1$ [cm$^{-3}$] & $1.5 \times 10^7$ & $3.5 \times 10^7$ & $7.7 \times 10^7$ & $1.8 \times 10^4$ & $1.8 \times 10^6$ & $9.0 \times 10^7$ & $4.0 \times 10^4$ & $3.0 \times 10^6$\\
  $n_1$ & 2.0 & 2.0 & 2.0 & 1.5 & 2.0 & 2.0 & 2.0 & 2.0\\
  $n_2$ & 3.5 & 3.5 & 3.5 & 3.5 & 4.1 & 3.0 & 3.5 & 2.5\\
  $\gamma_{\rm min}$ & $10^2$ & $10^3$ & $10^3$ & $10^3$ & $1$ & $3.0 \times 10^2$ & $10^3$ & $10^3$\\
  $\gamma_{\rm br}$ & $10^4$ & $10^4$ & $10^4$ & $10^4$ & $1.6 \times 10^3$ & $5.0 \times 10^2$ & $3.5 \times 10^5$ & $5.0 \times 10^4$\\
  $\gamma_c$ & $10^6$ & $10^7$ & $10^7$ & $10^7$ & $10^6$ & $4.0 \times 10^3$ & $6.0 \times 10^6$ & $10^6$\\
  \hline
\end{tabular}
\end{table}